Author's response to reviews

Title: Trends in prenatal care settings: Association with medical liability

Authors:

Andrew S Coco (ascoco@lancastergeneral.org)
Donna Cohen (docohen@lancastergeneral.org)
Michael A Horst (mahorst@lancastergeneral.org)
Angela S Gambler (sanders@lancastergeneral.org)

Version: 4 Date: 7 July 2009

Author's response to reviews:

July 7, 2009

Editorial Office
BioMed Central Journal
http://www.biomedcentral.com/
Tel: +44 (0)20 7631 9921
e-mail: editorial@biomedcentral.com

RE: Trends in Prenatal Care Settings: Association with Medical Liability,
MS 2142791766242056

Dear Editors,

Attached you will find the formatting changes, addressing the author list, affiliations, and table vertical elements as requested.

The formatting changes have been incorporated into the manuscript and the revised manuscript has been proofed again for typing and content errors.

Thank you again for your time, assistance, and consideration.

Sincerely,

Andrew Coco, MD, MS
Lancaster General Hospital
Health Research Institute
555 North Duke St, PO Box 3555
Lancaster, PA 17604